Author’s response to reviews

Title: Chyloptysis with chylopericardium, a rare case and mini-review

Authors:
XuMing Luo (303003671@qq.com)
Zhuhua Zhang (zhuzhu_19841007@126.com)
ShiQiang Wang (545706880@qq.com)
XianDong Gu (gu_213@hotmail.com)
Xiongbiao Wang (xiongbiao6@yahoo.com)

Version: 2 Date: 10 Jan 2018

Author’s response to reviews:

Thank you very much.

1. Please include a title page as the first page of your manuscript where you list the full names, institutional addresses and email addresses for all authors. The corresponding author should be indicated.

Answer: Thank you, it is corrected.

2. Please note that, according to Case Reports formatting, you have to include a section named 'Discussion and Conclusions', not only 'Conclusions'.

Answer: Thank you, it is corrected.

3. Please provide Table 1 only once.

Answer: Thank you, it is corrected.
4. Please, provide all the parts of Figure 2 as a unique file, and please provide unique citations for all the panels (we noted there are 2 panels depicted as Figure 2B).

Answer: Thank you, it is corrected.

5. Please amend your authors’ contribution section, as we noted that you provided these initials: LX, ZZ. Drafting the manuscript: BT, MS. Revisions for intellectual content: GS, WX., which do not match with the authors listed in the author list: WX, LX, ZZ, WS, GX. Please provide an Authors' contribution with all and only the authors listed.

Answer: Thank you, it is corrected.

6. Please remove the response to reviewers, as it is no longer necessary at this stage.

Answer: Thank you, it is corrected.

7. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Answer: OK, I do as indication.